

# 

**Citation:** Brasil P, Calvet GA, Siqueira AM, Wakimoto M, de Sequeira PC, Nobre A, et al. (2016) Zika Virus Outbreak in Rio de Janeiro, Brazil: Clinical Characterization, Epidemiological and Virological Aspects. PLoS Negl Trop Dis 10(4): e0004636. doi:10.1371/journal.pntd.0004636

Editor: Ann M Powers, Centers for Disease Control and Prevention, UNITED STATES

Received: February 6, 2016

Accepted: March 28, 2016

Published: April 12, 2016

**Copyright:** © 2016 Brasil et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

**Funding:** We thank the Conselho Nacional de Desenvolvimento Científico e Tecnológico – CNPq (grant no. 484735/2013-6 and 402135/2011-3), the Fundação de Amparo a Pesquisa no Estado do Rio de Janeiro-FAPERJ (grant no. E- 26/010.001558/ 2014) and the European Commission, grant FP7-21803 IDAMS (<u>http://www.idams.eu</u>), for having funded the study. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE** 

# Zika Virus Outbreak in Rio de Janeiro, Brazil: Clinical Characterization, Epidemiological and Virological Aspects

Patrícia Brasil<sup>1</sup>\*, Guilherme Amaral Calvet<sup>1</sup>, André Machado Siqueira<sup>1</sup>, Mayumi Wakimoto<sup>1</sup>, Patrícia Carvalho de Sequeira<sup>2</sup>, Aline Nobre<sup>3</sup>, Marcel de Souza Borges Quintana<sup>4</sup>, Marco Cesar Lima de Mendonça<sup>2</sup>, Otilia Lupi<sup>1</sup>, Rogerio Valls de Souza<sup>1</sup>, Carolina Romero<sup>1</sup>, Heruza Zogbi<sup>1</sup>, Clarisse da Silveira Bressan<sup>1</sup>, Simone Sampaio Alves<sup>2</sup>, Ricardo Lourenço-de-Oliveira<sup>5</sup>, Rita Maria Ribeiro Nogueira<sup>2</sup>, Marilia Sá Carvalho<sup>3</sup>, Ana Maria Bispo de Filippis<sup>2®</sup>, Thomas Jaenisch<sup>6®</sup>

1 Acute Febrile Illnesses Laboratory, Evandro Chagas National Institute of Infectious Diseases; Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 2 Flavivirus Laboratory, Oswaldo Cruz Institute/ Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 3 Scientific Computation Program, Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 4 Clinical Research Platform, Evandro Chagas National Institute of Infectious Diseases; Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 4 Clinical Research Platform, Evandro Chagas National Institute of Infectious Diseases; Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 5 Mosquito Transmitters of Hematozoans Laboratory, Oswaldo Cruz Institute/ Oswaldo Cruz Foundation (Fiocruz), Rio de Janeiro, Brazil, 6 Section Clinical Tropical Medicine, Department of Infectious Diseases, Heidelberg University Hospital, Heidelberg, Germany & German Center for Infectious Disease Research, Heidelberg, Germany

These authors contributed equally to this work.
\* patricia.brasil@ini.fiocruz.br

# Abstract

#### Background

In 2015, Brazil was faced with the cocirculation of three arboviruses of major public health importance. The emergence of Zika virus (ZIKV) presents new challenges to both clinicians and public health authorities. Overlapping clinical features between diseases caused by ZIKV, Dengue (DENV) and Chikungunya (CHIKV) and the lack of validated serological assays for ZIKV make accurate diagnosis difficult.

# Methodology / Principal Findings

The outpatient service for acute febrile illnesses in Fiocruz initiated a syndromic clinical observational study in 2007 to capture unusual presentations of DENV infections. In January 2015, an increase of cases with exanthematic disease was observed. Trained physicians evaluated the patients using a detailed case report form that included clinical assessment and laboratory investigations. The laboratory diagnostic algorithm included assays for detection of ZIKV, CHIKV and DENV. 364 suspected cases of Zika virus disease were identified based on clinical criteria between January and July 2015. Of these, 262 (71.9%) were tested and 119 (45.4%) were confirmed by the detection of ZIKV RNA. All of the samples with sequence information available clustered within the Asian genotype.

**Competing Interests:** The authors have declared that no competing interests exist.

#### **Conclusions / Significance**

This is the first report of a ZIKV outbreak in the state of Rio de Janeiro, based on a large number of suspected (n = 364) and laboratory confirmed cases (n = 119). We were able to demonstrate that ZIKV was circulating in Rio de Janeiro as early as January 2015. The peak of the outbreak was documented in May/June 2015. More than half of the patients reported headache, arthralgia, myalgia, non-purulent conjunctivitis, and lower back pain, consistent with the case definition of suspected ZIKV disease issued by the Pan American Health Organization (PAHO). However, fever, when present, was low-intensity and short-termed. In our opinion, pruritus, the second most common clinical sign presented by the confirmed cases, should be added to the PAHO case definition, while fever could be given less emphasis. The emergence of ZIKV as a new pathogen for Brazil in 2015 underscores the need for clinical vigilance and strong epidemiological and laboratory surveillance.

#### Author Summary

Zika virus (ZIKV) has been identified in 2015 in Brazil for the first time, causing outbreaks of an illness characterized by skin rash and absent or low grade and short-termed fever. It is difficult to distinguish ZIKV from Dengue (DENV) or (CHIKV) based on the acute clinical presentation. The virus is closely related to DENV, and therefore antibody tests also have problems distinguishing between the two viruses due to cross-reactivity. Recent findings suggest that in a minority of ZIKV cases neurological disease can develop, and that babies born from mothers reporting a ZIKV-like illness during pregnancy may suffer from congenital abnormalities, in many cases a small head or brain. Here we report about an outbreak of ZIKV disease in Rio de Janeiro in the first half of the year 2015, which reached its peak in May/June 2015. This is the first published description of a ZIKV outbreak in Latin America. It is interesting to note that confirmed cases appeared as early as January 2015. Cases were confirmed based on the detection of the viral genome in the blood of the patients. The clinical characterization of the confirmed cases and unconfirmed cases proved to be very similar. Itching or itching rash has been suggested to be added to the case definition issued by the Pan American Health Organization (PAHO).

# Introduction

#### Background

Since 2015, Brazil is faced with the challenge of three co-circulating arboviruses of major public health importance. For the last 30 years, dengue virus (DENV) infection has been the main mosquito-transmitted threat in the country, which has suffered several epidemics caused by all four serotypes, [1, 2] fostered by the widespread presence of its main vector, *Aedes aegypti*, in densely populated urban areas. [3-5] In spite of all efforts, a sustained reduction of the *Ae aegypti* population has not been achieved.

The emergence of Chikungunya virus (CHIKV) and Zika virus (ZIKV) in Brazil [6, 7] poses new challenges to clinicians and public health authorities due to overlapping clinical features and the fact that validated serological assays for ZIKV that can reliably distinguish between acute disease and past exposure are currently not available. Most importantly, ZIKV infections are suspected to be associated with congenital abnormalities [8] and with neurological complications such as Guillan-Barré syndrome (GBS) [9], while CHIKV infections have been associated with persisting arthralgia [10].

Both ZIKV as well as DENV are members of the family *Flaviviridae*, whereas CHIKV is an alphavirus of the *Togaviridae* family. ZIKV was first described in humans in Africa in 1952 [11] and has recently caused epidemics in the Pacific region. [12–14] In May 2015, ZIKV was identified in Northeast Brazil associated with an outbreak of acute exanthematous disease in the state of Bahia, [6, 15] followed by several other locations, including the state of Rio de Janeiro. [6, 15–17] ZIKV was considered to cause a benign infection, leading to a self-limited disease consisting of rash, fever (often of low intensity and short-termed or even absent), and mild arthralgia and therefore did not receive much scientific attention until recently [14].

However, new findings suggest an association of recent ZIKV disease with GBS amongst adults [18] and with congenital abnormalities (e.g. microcephaly and ocular lesions in neonates) born from women reporting a ZIKV-like disease during pregnancy. [19-21]

Here we describe the clinical, epidemiological, and virological features of patients with acute exanthematous disease at a tertiary reference centre in Rio de Janeiro during an outbreak of ZIVK in the first half of 2015. We analyze the temporal and spatial occurrence of suspected ZIKV patients over the time of the outbreak and compare the profile of clinical signs and symptoms in PCR-confirmed ZIKV patients with the current PAHO case definition for ZIKV disease.

#### Methods

#### Setting

The study was conducted at the Laboratorio de Pesquisa Clínica de Doenças Febris Agudas of the Instituto Nacional de Infectologia Evandro Chagas (INI), which is part of the regional Dengue-Research Center of Fundação Oswaldo Cruz (FIOCRUZ), a reference center of the dengue network in Rio de Janeiro, Brazil. Patients with acute febrile disease seen at this outpatient clinic are either referred from other health units in Rio de Janeiro, or are spontaneously seeking care as they live in the nearby district of 'Manguinhos'—an impoverished neighborhood close to the FIOCRUZ research center with a human development index of 0.726, a population of 36,160 persons in an area of 261 square kilometres, and an estimated number of 10,816 households.

Since January 2015 physicians at INI/FIOCRUZ observed an increase of cases characterized by rash, with either absent or low-grade and short-termed fever, and sometimes associated with arthralgia and/or conjunctivitis. The disease was clinically distinct from DENV which led to a systematic syndromic investigation utilizing a specific laboratory algorithm, which included diagnostics assays for detection of DENV, CHIKV, and ZIKV (after the reports of ZIKV transmission in the Northeast of Brazil[22]). The investigations were performed as part of the ongoing study on "Detection of unusual clinical presentations of dengue", which was reviewed and approved by the local Ethics Committee (CAAE 0026.0.009.000–07). Written informed consent was obtained from all patients or their legal representatives.

For the purpose of the analysis presented here we concentrate on the first half of the year 2015. After the identification of ZIKV infection in a patient with no travel history outside Rio de Janeiro in May 2015[16], additional prospective surveillance at the outpatients' clinic was initiated in order to specifically identify patients with suspected ZIKV disease. Patients with acute onset of generalized macular or papular rash were considered suspected cases of ZIKV. We interviewed patients using a standard case report form (CRF) to collect information about demographic features, clinical signs and symptoms, and the severity of the disease (S1 File). Blood samples were collected during the acute phase (i.e., within 7 days after the onset of

symptoms) and during the convalescent phase. Samples were stored at -70°C and analyzed in the Flavivirus Diagnostic and Reference Laboratory of Fiocruz.

#### Laboratory Analysis and Confirmation

Acute phase serum samples were tested by qRT-PCR for ZIKV [23] and RT-PCR for DENV RNA [24]. CHIKV qRT-PCR testing was performed for a random sample of 25% of patients as no patient tested positive and there was no on-going transmission of CHIKV in Rio at the time of this stud1y. Cases were classified as ZIKV infection if ZIKV RNA was detected in the serum.

# **Phylogenetic Analysis**

Phylogenetic analysis of nucleic acid sequences derived from 10 random ZIKV positive samples (out of 119) was performed using 327 base pairs of the envelope protein (GeneBank accession number KT381874). The tree was inferred using the maximum likelihood algorithm based on the Tamura 3-parameter model as implemented in MEGA 6. The numbers shown to the left of the nodes represent bootstrap support values > 70 (1,000 replicates). The tree was rooted with Spondoweni virus.

### Statistical Analysis

The home addresses of the cases were georeferenced using Google maps. Statistical analyses and maps were performed using software R version 3.2.2.libraries RCurl, RJSONIO, ggmap and ggplot2. [25]

#### Results

Between 1<sup>st</sup> of January and 31<sup>st</sup> of July 2015, 364 suspected cases of ZIKV disease were identified based on the presence of acute onset rash with or without fever. Of these, 262 (71.9%) were tested and 119 (45.4%) were confirmed by the detection of ZIKV RNA through qRT-PCR assay. All of the samples with sequence information available clustered within the Asian genotype (Fig 1). RT-PCR assays using consensus primers for nucleic acid of other arboviruses, including DENV and CHIKV, were negative in these patients. No DENV RNA was detected in any of the 143 acute phase serum samples that tested negative for ZIKV.

Suspected ZIKV cases started to appear in January 2015, peaking in May (Fig 2). Their geographical distribution is shown in Fig 3. In spite of the clustering in the surrounding neighborhood 'Manguinhos', cases residing in other neighborhoods in the Metropolitan area of Rio de Janeiro were also included (Fig 3).

Baseline characteristics of confirmed (tested ZIKV positive) and unconfirmed (tested ZIKV negative) patients are presented in <u>Table 1</u>. The median age of these patients was 37 years (range 9 to 60). A majority of the cases were females (158/262–60.3%), and among these 73% were aged 15–49 years (116/158–73.4%). The same pattern was seen with regard to the confirmed cases only (60.5% female [72/119] and 76.4% among those in the age group of 15–49 years [55/72]). No travel histories were recorded for the confirmed cases, thus infections were acquired locally. Only 38% of the patients recalled exposure to mosquito bites. Arterial hypertension was the most frequent comorbidity recorded, followed by diabetes. Geographic clustering of cases (defined as more than one case in the household, neighborhood or work) occurred in half of all confirmed cases (62/119–51.1%).

The most commonly reported symptoms in the first four days of the disease were rash, itching, prostration, headache and arthralgia (with or without associated oedema) (<u>Table 2</u>). Although fever was not observed at presentation in the majority of patients (64%), 43 patients





doi:10.1371/journal.pntd.0004636.g001

(36%) reported a history of fever not lasting longer than one day, or the occurrence of a single fever peak on the first day of disease. The median duration of rash was 5.5 days (range 3 to 7), and of arthralgia 9 days (range 2 to 21). Mild hemorrhagic symptoms (petechiae, minor mucosal bleeding) were reported in 21% of the confirmed and 11% of the unconfirmed cases, while no severe bleeding was observed. One patient was hospitalized with neurological symptoms consisting of peripheral nerve impairment, and is described in more detail elsewhere (Brasil et al, in press). No deaths or other severe complications were associated with the ZIKV disease in this series. Four women with confirmed ZIKV were pregnant. One had a miscarriage at 10 weeks of pregnancy, three weeks after the acute disease episode. The remaining three (two with ZIKV infection during the 18<sup>th</sup> week and one during the 35<sup>th</sup> week) delivered normal babies without any clinical evidence of infection or congenital abnormalities.

All patients had full blood counts performed. The median leucocyte count of confirmed ZIKV cases was 4,590 cells/mm<sup>3</sup> (2,240 to 11,570 cells/mm<sup>3</sup>), the median platelet count was 201,000 cells/mm<sup>3</sup> (102,000 to 463,000 cells/mm<sup>3</sup>) and the median haematocrit was 41.2% (33.2 to 50.3%).





doi:10.1371/journal.pntd.0004636.g002

#### Discussion

This is the first report of a ZIKV outbreak in the state of Rio de Janeiro, based on a large number of suspected (n = 364) and laboratory confirmed cases (n = 119). We were able to demonstrate that ZIKV was circulating in Rio de Janeiro as early as January 2015. The outbreak began in the first half of 2015 with a transmission peak in May 2015. The clinical picture of the disease was characterized by the sudden increase of exanthematous disease with absent or short-termed and low-grade fever.

The early documentation as well as the detailed clinical characterization was possible through a prospective syndromic surveillance study, which included a systematic collection of blood samples for the detection of unusual forms of dengue. In fact, 11% of the confirmed cases date back to the months before ZIKV transmission was reported in Northeast Brazil, in May (Fig 2).

The detection of ZIKV RNA in the serum of acutely ill patients and the absence of nucleic acid of other arboviruses provide convincing evidence that the outbreak was caused by ZIKV. Although DENV is common in Rio de Janeiro, none of the 252 patients for whom acute-phase specimens were available, tested positive for DENV, which is surprising and possibly highlights





doi:10.1371/journal.pntd.0004636.g003

explosive transmission dynamics of ZIKV. Patients were also tested for other pathogens at the attending physician's discretion, but no other causes were found (i.e. CHIKV, rubella, cytomegalovirus, EBV, toxoplasmosis). Based on this study, there is no evidence that ZIKV was circulating in Rio de Janeiro before January 2015. However, a larger and more representative sample would be necessary to make a more definitive statement about ZIKV circulation in Rio de Janeiro before 2015.

The peak of cases was observed in May (Fig 2), after the rainy season—which is in line with temporal patterns observed in previous outbreaks in Yap Island and French Polynesia. [13, 14] In May 2015, atypically high precipitation was recorded in Rio de Janeiro [26], which may have affected the vector distribution and abundance.

It is noteworthy, that the first DENV epidemic in Rio de Janeiro in 1986 started in the same geographical area from where the majority of ZIKV cases reported here originated [27]. This is likely to be associated with the high human population density, abundant *Ae aegypti* populations, the precarious socioeconomic status, and lack of infrastructure.

The observation that ZIKV cases clustered within households is interesting and needs to be verified in other settings. If this is a consistent feature, it could be consistent with a high

Patients Characteristics	Confirmed (n = 119) n(%)	Unconfirmed* (n = 143) n(%)	Total (n = 262) n(%)
Gender			
Male	47(40)	57(40)	104(40)
Female	72(61)	86(60)	158(60)
Age group			
8–14	4(3)	6(4)	10(4)
15–19	7(6)	6(4)	13(5)
20–29	28(24)	29(20)	57(22)
30–39	31(26)	43(30)	74(29)
40–49	25(21)	25(18)	50(19)
50–59	18(15)	21(15)	39(15)
> = 60	6(5)	13(9.)	19(7)
Comorbidities			
Hypertension	18(15)	20(14)	38(15)
Diabetes mellitus	6(7)	14(10)	20(8)
Asthma	4(3)	4(3)	8(3)
Smoking	11(9)	13(9)	24(9)
Regular alcohol comsumption	38(32)	36(25)	74(28)
Rhinitis	20(17)	21(15)	41(16)
Clustering	62(52)	65(46)	127(49)
Household	28(45)	25(39)	53(42)
Neighborhood	20(32)	19(29)	39(31)
Work	14(23)	21(32)	35(28)
Insect bite	48(40)	50(35)	98(38)

#### Table 1. Baseline characteristics of confirmed and unconfirmed ZIKV disease patients.

\* tested ZIKV negative

doi:10.1371/journal.pntd.0004636.t001

vectorial capacity or hint towards other routes of transmission. The possibility of transmission via mucosal contact is supported by evidence of sexual transmission [28, 29] and by the detection of ZIKV RNA in saliva and urine. [30, 31] Further studies need to be carried out to address this question in more detail, with important implications on epidemiology and disease control of ZIKV.

The age distribution with a bias towards adults is consistent with the overall age profile of the clinic. Female patients were overrepresented with regard to males, which might be due to differences in health care-seeking behavior.

An epidemic of microcephaly in newborns in Northeast Brazil has been the focus of attention by public health authorities in Brazil and worldwide. There is emerging evidence that these congenital abnormalities may be associated with exposure to ZIKV during pregnancy. [32] Among the ZIKV-positive pregnant women followed in our study, one out of four had a miscarriage in the 10<sup>th</sup> week of pregnancy. However, no further investigations were performed in this case. In a recently published cohort of pregnant women with ZIKV infection, we have described two miscarriages amongst 72 confirmed cases [21], although the real frequency of this and other complications is yet to be determined. More recently, the state of Rio de Janeiro has reported a tenfold increase in microcephaly cases in 2015 compared to previous years [32] and, due to the long-lasting medical and economic consequences, it is of high importance that multidisciplinary studies are conducted to determine the incidence and the phenotypic variability of congenital abnormalities related to ZIKV infection during pregnancy.

#### Table 2. Clinical characteristics of confirmed and unconfirmed\* ZIKV infected patients, in the first four days of disease.

Sign/Symptom	Confirmed n(%)	Unconfirmed n(%)
Macular or papular rash	115(97)	113(79)
Itching	94(79)	127(89)
Prostration	87(73)	104(73)
Headache	78(66)	101(71)
Arthralgia	75(63)	105(73)
Myalgia	73(61)	96(67)
Nonpurulent conjuntivitis	66(56)	57(40)
Low back pain	61(51)	70(49)
Retro-orbital pain	53(45)	76(53)
Lymphonodes elargement	49(41)	33(23)
Chills	44(37)	52(36)
Fever	43(36)	71(50)
Anorexia	42(35)	60(42)
Photophobia	41(345)	43(30)
Oropharyngeal pain	38(32)	35(25)
Edema	34(29)	26(18)
Taste alteration	32(27)	33(23)
Nausea	28(24)	43(30)
Enanthema/Petechiae/Bleeding	25(21)	16(11)
Nasal congestion	24(20)	25(18)
Sweating	23(19)	38(27)
Diarrhea	23(19)	21(15)
Abdominal pain	20(17)	29(20)
Cough	19(16)	28(20)
Coryza	18(15)	20(14)
Lipothymia	18(15)	28(20)
Hoarseness	13(11)	13(9)
Earache	11(9)	8(5)
Dysuria	8(7)	5(4)
Choluria	7(6)	8(6)
Dyspnea	7(6)	10(7)
Vomiting	5(4)	10(7)
Hepatomegaly	2(2)	2(1)

\* tested ZIKV negative

doi:10.1371/journal.pntd.0004636.t002

The clinical signs and symptoms of ZIKV observed in Rio de Janeiro have many similarities with those described in previous reports and case series. [14] The similarity and frequency of clinical manifestations between confirmed and unconfirmed cases supports the notion that the majority of suspect cases were suffering from ZIKV disease, which could be due to current limits of detecting low quantities of RNA within a potentially short viremic period and the lack of a reliable serological test to ascertain recent exposure and distinguish it from DENV infections.

As in DENV and CHIKV infections, rash was a common feature in ZIKV. However, pruritus/itching was a prominent characteristic of the maculo-papular rash in the majority of confirmed cases (79%) from the onset of symptoms, which can potentially help in distinguishing ZIKV from other arboviral infections. Enlarged lymph nodes, especially in the cervical and retro-auricular regions, were also found frequently, which led clinicians to consider rubella as an important differential diagnosis. More than half of the patients reported headache, arthralgia, myalgia, non-purulent conjunctivitis, and lower back pain, consistent with the case definition of suspected ZIKV issued by PAHO. [8] However, fever was absent in most cases. In our opinion, pruritus, the second most common clinical sign presented by the confirmed cases, should be added to the PAHO case definition.

DENV, which has been circulating in Brazil for many years, was considered an unlikely diagnosis in many patients because fever was only observed in a minority (36%). When present, fever was usually short-termed and the temperature relatively low. From a clinical management perspective, it is important to differentiate ZIKV both from CHIKV and especially from DENV due to the need to monitor cases for possible evolution of severe and life-threatening clinical outcomes. Standard laboratory values (e.g. hematology or biochemistry) did not show a distinct pattern in the confirmed ZIKV patients and are unlikely to be helpful. The leucocyte count in ZIKV patients was in many cases moderately decreased as in other viral infections (< 5,000 cells mm<sup>3</sup>) and both platelet counts and haematocrit were within the normal range.

In this study, only one patient was hospitalized with febrile illness and a neurological manifestation developed afterwards (manuscript submitted). An increase in cases with Gullain-Barré syndrome (GBS) was reported during the epidemic In French Polynesia in 2013/14 [12], as well as in the Northeast of Brazil during the current outbreak. [6] It remains to be seen if ZIKV is causally related to higher rates of neurological complications.

Our data suggests that in a previously naïve adult population, ZIKV causes a self-limited and mostly benign disease characterized by a pruritic maculopapular rash and absent or lowgrade, short-termed fever. ZIKV-like disease was not a notifiable condition in Rio de Janeiro until October 22th, 2015. This partially explains why our data differs from the official figures, [22] as the syndromic surveillance study that was implemented in 2007 (long before the ZIKV outbreak started) is more likely to detect changes in the clinical patterns of dengue compared to the routine health system. Similar to the dynamic in the Northeast of Brazil, a considerable increase in the number of microcephaly in neonates was registered in Rio de Janeiro during 2015. As of January 2016, 122 microcephaly cases were documented—compared to the average of 10 cases per year registered in the previous years. [32] This increase could potentially be attributed to the first wave of ZIKV transmission observed between January and July 2015 in Rio de Janeiro. In Pernambuco, in the Northeast region of Brazil, 1306 cases of microcephaly were reported as of the second epidemiological week of 2016 [32] which are attributed to the ZIKV outbreak in early spring 2015.

At this point, health services must be alerted to the potential for an even larger epidemic during the summer of 2015–2016 spreading to additional locations and affecting the susceptible proportion of the population that was not exposed during the last transmission season.

The emergence of ZIKV as a new pathogen for Brazil in 2015 underscores the ease with which pathogens travel between continents and the need for clinical vigilance and strong epidemiological and laboratory surveillance systems. The phylogenetic analysis somehow is in line with the speculative hypothesis that ZIKV was possibly introduced to Rio de Janeiro during the VI World Sprint Championship canoe race in August 2014, which included teams from four Pacific countries (French Polynesia, New Caledonia, Cook Islands, and Easter Island) where the virus circulated during 2014. [33]

In 2016, Rio de Janeiro will be hosting the Summer Olympics and Paralympics games, which will attract a high number of national and international visitors. The threat caused by ZIKV has far reaching implications on tourism and industry.[<u>34</u>] Reliable and sensitive surveillance of arboviral disease that includes a system for the detection of emerging pathogens is of paramount importance to manage the complex challenges ahead. Our findings have demonstrated that ZIKV was circulating in Rio de Janeiro at least five months before its detection was

announced by the health authorities, which must be taken into consideration for future design and implementation of effective syndromic surveillance systems.

#### **Supporting Information**

**S1 File. Standard case report form.** (PDF)

#### **Author Contributions**

Conceived and designed the experiments: PB. Performed the experiments: PB GAC AMS AN MdSBQ OL RVdS CR HZ CdSB PCdS MCLdM. Analyzed the data: GAC MW. Contributed reagents/materials/analysis tools: PCdS MCLdM SSA RMRN AMBdF. Wrote the paper: PB GAC AMS MW RLdO MSC AMBdF TJ.

#### References

- Villabona-Arenas CJ, de Oliveira JL, Capra Cde S, Balarini K, Loureiro M, Fonseca CR, et al. Detection of four dengue serotypes suggests rise in hyperendemicity in urban centers of Brazil. PLoS neglected tropical diseases. 2014; 8(2):e2620. Epub 2014/03/04. doi: <u>10.1371/journal.pntd.0002620</u> PMID: <u>24587454</u>; PubMed Central PMCID: PMC3937275.
- Fares RC, Souza KP, Anez G, Rios M. Epidemiological Scenario of Dengue in Brazil. BioMed research international. 2015; 2015:321873. Epub 2015/09/29. doi: <u>10.1155/2015/321873</u> PMID: <u>26413514</u>; PubMed Central PMCID: PMC4568054.
- Araujo HR, Carvalho DO, Ioshino RS, Costa-da-Silva AL, Capurro ML. Aedes aegypti Control Strategies in Brazil: Incorporation of New Technologies to Overcome the Persistence of Dengue Epidemics. Insects. 2015; 6(2):576–94. Epub 2015/10/16. doi: <u>10.3390/insects6020576</u> PMID: <u>26463204</u>; PubMed Central PMCID: PMC4553499.
- Favaro EA, Dibo MR, Pereira M, Chierotti AP, Rodrigues-Junior AL, Chiaravalloti-Neto F. Aedes aegypti entomological indices in an endemic area for dengue in Sao Paulo State, Brazil. Revista de saude publica. 2013; 47(3):588–97. Epub 2013/12/19. PMID: <u>24346572</u>.
- Codeco CT, Honorio NA, Rios-Velasquez CM, Santos Mda C, Mattos IV, Luz SB, et al. Seasonal dynamics of Aedes aegypti (Diptera: Culicidae) in the northernmost state of Brazil: a likely port-of-entry for dengue virus 4. Memorias do Instituto Oswaldo Cruz. 2009; 104(4):614–20. Epub 2009/09/02. PMID: 19722086.
- Cardoso CW, Paploski IA, Kikuti M, Rodrigues MS, Silva MM, Campos GS, et al. Outbreak of Exanthematous Illness Associated with Zika, Chikungunya, and Dengue Viruses, Salvador, Brazil. Emerging infectious diseases. 2015; 21(12):2274–6. Epub 2015/11/20. doi: <u>10.3201/eid2112.151167</u> PMID: <u>26584464</u>; PubMed Central PMCID: PMC4672408.
- Nunes MR, Faria NR, de Vasconcelos JM, Golding N, Kraemer MU, de Oliveira LF, et al. Emergence and potential for spread of Chikungunya virus in Brazil. BMC Med. 2015; 13:102. doi: <u>10.1186/s12916-015-0348-x</u> PMID: <u>25976325</u>; PubMed Central PMCID: PMC4433093.
- PAHO/WHO 1 December 2015. Epidemiological Alert Neurological syndrome, congenital malformations, and Zika virus infection. Implications for public health in the Americas. Available at: <a href="http://www.paho.org/hq/index.php?option=com\_docman&task=doc\_download&Itemid=&gid=32405&Iang=en">http://www.paho.org/hq/index.php?option=com\_docman&task=doc\_download&Itemid=&gid=32405&Iang=en</a>. Accessed 16.12.2015.
- Oehler E, Watrin L, Larre P, Leparc-Goffart I, Lastere S, Valour F, et al. Zika virus infection complicated by Guillain-Barre syndrome—case report, French Polynesia, December 2013. Euro surveillance: bulletin Europeen sur les maladies transmissibles = European communicable disease bulletin. 2014; 19(9). Epub 2014/03/15. PMID: <u>24626205</u>.
- Pialoux G, Gauzere BA, Jaureguiberry S, Strobel M. Chikungunya, an epidemic arbovirosis. The Lancet infectious diseases. 2007; 7(5):319–27. Epub 2007/04/24. doi: <u>10.1016/S1473-3099(07)70107-X</u> PMID: <u>17448935</u>.
- 11. Simpson DI. Zika Virus Infection in Man. Trans R Soc Trop Med Hyg. 1964; 58:335–8. PMID: 14175744.
- Musso D, Nilles EJ, Cao-Lormeau VM. Rapid spread of emerging Zika virus in the Pacific area. Clin Microbiol Infect. 2014; 20(10):O595–6. doi: <u>10.1111/1469-0691.12707</u> PMID: <u>24909208</u>.

- Ioos S, Mallet HP, Leparc Goffart I, Gauthier V, Cardoso T, Herida M. Current Zika virus epidemiology and recent epidemics. Medecine et maladies infectieuses. 2014; 44(7):302–7. doi: <u>10.1016/j.medmal.</u> <u>2014.04.008</u> PMID: <u>25001879</u>.
- Duffy MR, Chen TH, Hancock WT, Powers AM, Kool JL, Lanciotti RS, et al. Zika virus outbreak on Yap Island, Federated States of Micronesia. N Engl J Med. 2009; 360(24):2536–43. doi: <u>10.1056/</u> NEJMoa0805715 PMID: 19516034.
- Campos GS, Bandeira AC, Sardi SI. Zika Virus Outbreak, Bahia, Brazil. Emerging infectious diseases. 2015; 21(10):1885–6. Epub 2015/09/25. doi: <u>10.3201/eid2110.150847</u> PMID: <u>26401719</u>; PubMed Central PMCID: PMC4593454.
- Calvet GA, Filippis AMB, Mendonça MCL, Sequeira PC, Siqueira AM, Veloso VG, et al. First detection of autochthonous Zika virus transmission in a HIV-infected patient in Rio de Janeiro, Brazil. Journal of Clinical Virology. 2016; 74:1–3. Epub 2015/11/30. doi: 10.1016/j.jcv.2015.11.014 PMID: 26615388.
- Zanluca C, de Melo VC, Mosimann AL, Dos Santos GI, Dos Santos CN, Luz K. First report of autochthonous transmission of Zika virus in Brazil. Memorias do Instituto Oswaldo Cruz. 2015; 110(4):569– 72. Epub 2015/06/11. doi: <u>10.1590/0074-02760150192</u> PMID: <u>26061233</u>; PubMed Central PMCID: PMC4501423.
- Brito C. Zika Virus: A New Chapter in the History of Medicine. Acta Medica Portugesa. 2015; 28 (6):679–80.
- Besnard M, Mallet HP: Increase of cerebral congenital malformations among newborns and fetus in French Polynesia, 2014–2015, following a Zika virus outbreak. <u>https://m.box.com/shared\_item/https%</u> <u>3A%2F%2Fapp.box.com%2Fs%2Fptvqdargwr10m3ag849rv59c5iditco7</u>.
- de Paula Freitas B, de Oliveira Dias JR, Prazeres J, Sacramento GA, Ko AI, Maia M, et al. Ocular Findings in Infants With Microcephaly Associated With Presumed Zika Virus Congenital Infection in Salvador, Brazil. JAMA Ophthalmol. 2016. doi: <u>10.1001/jamaophthalmol.2016.0267</u> PMID: <u>26865554</u>.
- Brasil P, Pereira JP Jr., Raja Gabaglia C, Damasceno L, Wakimoto M, Ribeiro Nogueira RM, et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro—Preliminary Report. N Engl J Med. 2016. doi: <u>10.</u> 1056/NEJMoa1602412 PMID: 26943629.
- Brazil Ministry of Health press release. Available at: <u>http://portalsaude.saude.gov.br/index.php/</u> cidadao/principal/agencia-saude/17701-confirmacao-do-zika-virus-no-brasil. Accessed 19.01.2016.
- Lanciotti RS, Kosoy OL, Laven JJ, Velez JO, Lambert AJ, Johnson AJ, et al. Genetic and serologic properties of Zika virus associated with an epidemic, Yap State, Micronesia, 2007. Emerg Infect Dis. 2008; 14(8):1232–9. doi: <u>10.3201/eid1408.080287</u> PMID: <u>18680646</u>; PubMed Central PMCID: PMC2600394.
- Lanciotti RS, Calisher CH, Gubler DJ, Chang GJ, Vorndam AV. Rapid detection and typing of dengue viruses from clinical samples by using reverse transcriptase-polymerase chain reaction. Journal of clinical microbiology. 1992; 30(3):545–51. Epub 1992/03/01. PMID: <u>1372617</u>; PubMed Central PMCID: PMC265106.
- 25. R Core Team (2015). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL <u>https://www.R-project.org/</u>.
- Instituto Nacional de Meteorologia (INMET). Available at: <u>http://www.inmet.gov.br/sonabra/pg\_iframe.php?codEst=A652&mesAno=2015</u>. Accessed 16.12.2015.
- Schatzmayr HG, Nogueira RM, Travassos da Rosa AP. An outbreak of dengue virus at Rio de Janeiro —1986. Mem Inst Oswaldo Cruz. 1986 Apr-Jun; 81(2):245–6. PMID: 3587006
- Musso D, Roche C, Robin E, Nhan T, Teissier A, Cao-Lormeau VM. Potential sexual transmission of Zika virus. Emerg Infect Dis. 2015; 21(2):359–61. doi: <u>10.3201/eid2102.141363</u> PMID: <u>25625872</u>; PubMed Central PMCID: PMC4313657.
- Foy BD, Kobylinski KC, Chilson Foy JL, Blitvich BJ, Travassos da Rosa A, Haddow AD, et al. Probable non-vector-borne transmission of Zika virus, Colorado, USA. Emerging infectious diseases. 2011; 17 (5):880–2. Epub 2011/05/03. doi: <u>10.3201/eid1705.101939</u> PMID: <u>21529401</u>; PubMed Central PMCID: PMC3321795.
- Gourinat AC, O'Connor O, Calvez E, Goarant C, Dupont-Rouzeyrol M. Detection of Zika virus in urine. Emerging infectious diseases. 2015; 21(1):84–6. Epub 2014/12/23. doi: <u>10.3201/eid2101.140894</u> PMID: 25530324; PubMed Central PMCID: PMC4285245.
- Musso D, Roche C, Nhan TX, Robin E, Teissier A, Cao-Lormeau VM. Detection of Zika virus in saliva. J Clin Virol. 2015; 68:53–5. doi: 10.1016/j.jcv.2015.04.021 PMID: 26071336.
- Brazil. Ministry of Health. Microcephaly. <u>http://portalsaude.saude.gov.br/index.php/cidadao/principal/agencia-saude/21254-ministerio-da-saude-atualiza-numeros-de-microcefalia-relacionados-ao-zika</u>. Accessed 14.01.2016.

- Musso D. Zika Virus Transmission from French Polynesia to Brazil. Emerging infectious diseases. 2015; 21(10):1887. Epub 2015/09/26. doi: <u>10.3201/eid2110.151125</u> PMID: <u>26403318</u>; PubMed Central PMCID: PMC4593458.
- Bogoch II, Brady OJ, Kraemer MU, German M, Creatore MI, Kulkarni MA, et al. Anticipating the international spread of Zika virus from Brazil. Lancet. 2016; 387(10016):335–6. Epub 2016/01/19. doi: <u>10.</u> <u>1016/S0140-6736(16)00080-5</u> PMID: <u>26777915</u>.